Title: Preventing disease through opportunistic, Rapid EngagEMent by Primary Care Teams using Behaviour Change Counselling. (PRE-EMPT): protocol for a general practice-based cluster randomised trial.

Authors:

Clio Spanou (spanouc@cardiff.ac.uk)
Sharon A Simpson (simpsonsa@cf.ac.uk)
Kerry Hood (hoodk1@cf.ac.uk)
Adrian Edwards (edwardsag@cf.ac.uk)
David Cohen (dcohen@glam.ac.uk)
Stephen Rollnick (rollnick@cf.ac.uk)
Ben Carter (b.r.carter@bham.ac.uk)
Jim McCambridge (Jim.McCambridge@lshtm.ac.uk)
Laurence Moore (MooreL1@cf.ac.uk)
Elizabeth Randell (Randelle@cf.ac.uk)
Timothy Pickles (PicklesTE@cf.ac.uk)
Christine Smith (smithc5@cf.ac.uk)
Claire Lane (cal968@bhma.ac.uk)
Fiona Wood (WOOD@cf.ac.uk)
Hazel Thornton (hazelcagct@keme.co.uk)
Chris C Butler (butlercc@cf.ac.uk)

Version: 2 Date: 21 September 2010

Author's response to reviews:

According to feedback received on Major revisions:

• The PreEmpt Study Team has been replaced with the actual names of the members both in the title and in the authors contributions section.
• References are now cited in the text using consecutive numbers in square brackets and placed before the full stop.

Minor revisions made:

• The formatting of the Title has been revised as advised.
• Author initials are now used to link each author to their corresponding email address.
• full citation details for references 2 and 59 are provided
• DOI number is removed from references 44 and 60.
• The manuscript is checked for any typographical errors and any final corrections or revisions were made as appropriate.